Author’s response to reviews

Title: Neurophysiologic effects of spinal manipulation in patients with chronic low back pain

Authors:

Brian C Clark (clarkb2@ohio.edu)
David A Goss (dg18080@ohio.edu)
Stevan Walkowski (walkowski@oucom.ohiou.edu)
Richard L Hoffman (hoffman@oucom.ohiou.edu)
Andrew Ross (aross.635@gmail.com)
James S Thomas (thomasj5@ohio.edu)

Version: 4 Date: 13 July 2011

Author’s response to reviews: see over
July 13th, 2011

Re: MS: 2045048865354825

Dear BMC Musculoskeletal Disorders,

As per the editorial teams request we have made corrections to all of the formatting issues listed below, except the one relating to the white spacing of the figures. As was noted, we are under the assumption that the editorial team can make these changes and we are confident that they will perform this task more adequately than we will be able to.

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

Abstract: Please change 'Summary of background data' to 'Background'.

References: Please make sure references are cited in numerical order.

Additional files: Please remove additional files from manuscript.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Line numbers: Please remove these.

Author list: Please remove titles.

Highlighting/tracking: Please remove all highlighting/tracking from the manuscript.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Please let me know if you have any questions.

Sincerely,

Brian C. Clark, Ph.D.
Executive Director, Ohio Musculoskeletal & Neurological Institute (OMNI)
Associate Professor of Physiology, Department of Biomedical Sciences
Ohio University Heritage College of Medicine